Author's response to reviews

Title: Behavioural interventions for weight management in pregnancy: A systematic review of quantitative and qualitative data.

Authors:

Fiona Campbell (f.campbell@sheffield.ac.uk)
Maxine Johnson (m.johnson@sheffield.ac.uk)
Josie Messina (j.messina@sheffield.ac.uk)
Louise Guillaume (l.guillaume@sheffield.ac.uk)
Elizabeth Goyder (e.goyder@sheffield.ac.uk)

Version: 3 Date: 28 April 2011

Author's response to reviews:

In addition to the amendments made following the reviewers comments I have made all the changes requested by the editorial production team. These include: removing track changes, revising the abstract so it has the correct structure, changing the structure of the paper ensuring the correct headings, removed reference 39, ensured the tables are referenced in the text, listed the figures, added these as additional files and removed them from the text, removed vertical lines from the tables.